Author's response to reviews

Title: Exposure to cooking fuels and birth weight in Lanzhou, China: a birth cohort study

Authors:

Min Jiang (athene.jiang@hotmail.com)
Jie Qiu (qioujie@21cn.com)
Min Zhou (ghz66383399@sina.com)
Xiaochun He (LZxiaoachunhe@126.com)
Hongmei Cui (369678469@qq.com)
Catherine Lerro (catherine.lerro@yale.edu)
Ling Lv (1358638351@qq.com)
Xiaojuan Lin (jxlinmail@163.com)
Chong Zhang (zhch1972@163.com)
Honghong Zhang (64276995@qq.com)
Ruifeng Xu (xrf730130@163.com)
Daling Zhu (zdlldz@126.com)
Yun Dang (34370100@qq.com)
Xudong Han (hxd809904@126.com)
Hanru Zhang (zhanghrfred@163.com)
Haiya Bai (17993902@qq.com)
Ya Chen (cy704@163.com)
Zhongfeng Tang (tzf3325@gmail.com)
Ru Lin (1632624248@qq.com)
Tingting Yao (yaott821005@163.com)
Jie Su (50249405@qq.com)
Xiaoying Xu (794983177@qq.com)
Xiaohui Liu (lxhzxq609@163.com)
Wendi Wang (xiaodismile@hotmail.com)
Yueyuan Wang (wangyueyuan0618@163.com)
Bin Ma (153873545@qq.com)
Weitao Qiu (bjbqwt@qq.com)
Cairong Zhu (cairong.zhu@hotmail.com)
Suping Wang (spwang88@163.com)
Huang Huang (h.huang@yale.edu)
Nan Zhao (nan.zhao@yale.edu)
Xiaosong Li (lixiaosong1101@126.com)
Qing Liu (2305470816@qq.com)
Yawei Zhang (yawei.zhang@yale.edu)

Version: 2  Date: 5 March 2015

Author's response to reviews:

Dear Editor,
The following changes have been made on the manuscript.
(1) Line and page numbers have been included.
(2) Authors' Contributions section has been placed after the Conflicts of Interest Statement section.
(3) Tables have been moved after the references.

Thank you for considering our manuscript.

Sincerely yours,

Yawei Zhang, M.D., Ph.D., M.P.H.
Associate Professor
Yale University School of Public Health
60 College Street, LEPH 440
203-7856210 (phone)
203-7376023 (fax)
yawei.zhang@yale.edu (email)